Author’s response to reviews

Title: Multi-parametric neuroimaging evaluation of cerebrotendinous xanthomatosis and their correlation with the neuropsychological presentations

Authors:

Chiung-Chih Chang (neur099@adm.cgmh.org.tw)
Chun-Chung Lui (lchung@adm.cgmh.org.tw)
Jiun-Jie Wang (jiunjie.wang@gmail.com)
Shu-Hua Huang (sophia4790@adm.cgmh.org.tw)
Cheng- Hsien Lu (Chlu99@ms44.url.com.tw)
Ching Chen (tuku633@yahoo.com.tw)
Chih-Feng Chen (xfroggykimo@yahoo.com.tw)
Min-Chien Tu (tmctmc@adm.cgmh.org.tw)
Chi-Wei Huang (justin1124@adm.cgmh.org.tw)
Wen-Neng Chang (cwenneng@ms19.hinet.net)

Version: 4 Date: 29 June 2010

Author’s response to reviews:

Dear BMC neurology editorial,

Thank you for your feedback. We had provided the major and minor revision accordingly.

For the authorship orders, we checked the manuscript and the submission system, both were in the following orders: Chiung-Chih Chang, Chun-Chung Lui, Jiun-Jie Wang, Shu-Hua Huang, Cheng- Hsien Lu, Ching Chen, Chih-Feng Chen, Min-Chien Tu, Chi-Wei Huang and Wen-Neng Chang.

Please do not hesitate to contact with me if there is any further query.

Chang Wen-Neng